SYSTEMATIC REVIEW

www.jvsgbi.com

Safety and efficacy of tranexamic acid in major non-cardiac vascular surgery: a systematic review and meta-analysis

Atha K,¹ Corrigan L,¹ Bera K,^{2*} Shah A^{3,4,5*}

* Contributed equally

- 1. Medical Student, University of Oxford Medical School, Oxford, UK
- Senior Clinical Fellow, Department of Vascular Surgery, Oxford University Hospitals NHS Foundation Trust, Oxford, UK
- Senior Clinical Research Fellow, Nuffield Department of Clinical Neurosciences, University of Oxford, Oxford, UK
- NIHR Blood and Transplant Research Unit in Data Driven Transfusion Practice, Radcliffe, Department of Medicine, University of Oxford, Oxford, UK
- 5. Consultant Anaesthetist, Department of Anaesthesia, Hammersmith Hospital, Imperial College Healthcare NHS Trust, London, UK

Corresponding author:

Akshay Shah Nuffield Department of Clinical Neuroscience, Level 6 West Wing, John Radcliffe Hospital, Oxford OX3 9DU, UK Email: akshay.shah@ ndcn.ox.ac.uk

Received: 21st June 2024 Accepted: 7th August 2024 Online: 26th August 2024

Plain English Summary

Why we undertook the work: Tranexamic acid (TXA) is a medicine used to reduce bleeding in people undergoing major surgery by stabilising blood clots. However, too much clot can obstruct blood flow to important organs such as the heart, lungs and brain. There is a delicate balance between the formation and breakdown of clots. Using drugs such as TXA can help reduce bleeding and avoid the need for blood transfusion, but there are some concerns that these clots may cause harm by limiting blood flow to important organs. So far, TXA has been shown to reduce bleeding without causing clot-associated problems in patients having open heart surgery, women giving birth and in those who are injured from trauma. However, TXA has not been used very much in vascular surgery (surgery on major blood vessels of the body) because the procedures (which include applying clamps to stop blood flow) are per se more prone to clot formation. The aim of this research was to assess the safety and benefits of TXA in people having vascular surgery.

What we did: We searched seven large research databases for studies that looked at TXA versus a placebo (dummy treatment) or no TXA. We were interested in studies that looked at adult patients. We studied and summed up the results of these studies.

What we found: We found three studies. These included a total of 1,560 adults, which compared TXA to normal saline (salty water) in patients having vascular surgery. In two studies the TXA was given in a vein and in one study it was given as tablets. We found that TXA given to people through a drip does not increase the risk of clot-related problems such as heart attacks or blood clots in the legs. On the other hand, we found no evidence that TXA reduced severe bleeding (eg, major organ bleeds, fatal bleeds) either. However, our confidence in these results is low. One study investigated the use of TXA tablets in patients who had their aorta repaired. The use of TXA for 30 days in these patients had no effect on blood leakage around the repair site. This study also did not find that TXA increased the number of clot-related problems.

What this means: The information from our research suggests that TXA does not increase the risk of clot-related problems in patients having vascular surgery. However, the current studies are limited by small numbers of patients. This means that we do not have enough information to say for certain whether or not we should use TXA in vascular surgery. Ongoing and future studies will help reduce this uncertainty and provide more definitive information for doctors and patients.

Abstract

Background: Tranexamic acid (TXA) is a synthetic lysine analogue that inhibits fibrinolysis. The effectiveness of TXA in obstetrics, trauma and orthopaedic and cardiac surgery is well established. However, concerns regarding its prothrombotic potential remain, which is an important consideration for vascular surgery. We aimed to evaluate the safety and efficacy of TXA in adults undergoing major non-cardiac vascular surgery.

Methods: A pre-specified protocol (PROSPERO CRD42023427282) was followed. Relevant databases (PubMed, MEDLINE, EMBASE) were searched for randomised controlled trials (RCTs). Data extraction and risk of bias assessments were performed in duplicate. A random effects model was used to synthesise data from RCTs. Measures of effect were reported as relative risk (RR) with 95% confidence intervals (CIs). The primary safety outcome was a composite of arterial and venous thromboembolic events (composite of myocardial infarction, non-haemorrhagic stroke, peripheral arterial thrombosis and symptomatic proximal venous thromboembolism). Certainty of evidence was assessed using the GRADE approach.

Results: After screening 1989 records, three RCTs were included, cumulatively enrolling 1,560 participants. In all trials, patients received either TXA (intravenously or orally) compared with placebo. There was no evidence of the effect of intravenous TXA on thromboembolic events (RR 1.10, 95% CI 0.89 to 1.36, two RCTs, 1460 participants, low certainty of evidence) or on critical bleeding (composite of life-threatening, critical and major organ bleed) (RR 0.85, 95% CI 0.65 to 1.11, two RCTs, 1460 participants, low certainty of evidence). TXA may reduce postoperative blood loss, especially at 0–4 hours (Cliff's delta –0.41, 95% CI –0.19 to –0.59) and 0–24 hours (Cliff's delta –0.37, 95% CI –0.14 to –0.55) after surgery. There was no evidence of an effect of TXA on reducing perioperative red blood cell (RBC) transfusion requirements (RR 0.66, 95% CI 0.11 to 3.95, one RCT, 100 participants). One RCT assessed oral TXA and found no evidence of an effect on type II endoleak post endovascular aneurysm repair. No thrombotic complications were reported in this RCT during the study period.

Conclusions: We found no evidence of an effect of TXA on thromboembolic complications. While TXA appears to reduce early postoperative bleeding, the clinical relevance of this is uncertain. Due to limitations of study design and the variety of vascular procedures, the role of TXA in vascular surgery is still unclear. Ongoing trials may reduce this uncertainty.

Key words: tranexamic acid, vascular surgery, thromboembolism blood transfusion, bleeding

Registration: Prospectively registered on PROSPERO (CRD42023427282)

Introduction

Tranexamic acid (TXA) is a synthetic lysine analogue that inhibits the conversion of plasminogen to plasmin, thereby inhibiting fibrinolysis.¹ Large pragmatic randomised controlled trials (RCTs) have demonstrated that TXA is associated with a reduction in mortality in patients with major traumatic haemorrhage,² postpartum haemorrhage³ and mild to moderate traumatic brain injury⁴; reduced transfusion requirements in adult and paediatric cardiac surgery^{5,6}; and caesarean delivery⁷; and fewer critical bleeding events in major non-cardiac surgery.⁸ TXA is on the World Health Organization's list of essential medicines.⁹

Despite evidence of efficacy in these settings, owing to the mechanism of action of TXA, concerns about thromboembolic risk remain. This risk may be more pertinent in patients with active thromboembolic disease, inappropriate administration or those with unbalanced haemostatic systems favouring thrombosis. In a large RCT enrolling patients with gastrointestinal bleeding, TXA was associated with an increased risk of venous thrombosis with no effect on mortality.¹⁰ Possible reasons for these findings include the high dose of TXA administered (4g over 24 hours) and the delayed presentation to hospital of many of the patients, missing the early period of excessive fibrinolysis.

Patients undergoing major non-cardiac vascular surgery are per se more prone to thrombosis than general surgery patients.^{1,11} Hypercoagulability caused by TXA can potentially cause damage to – and occlude arteries recently operated on – new bypass grafts or endovascular stents and stent grafts, which can lead to rare but devastating complications such as limb loss and even death.¹² Although systematic reviews have not demonstrated an increased risk of thromboembolic complications associated with TXA in surgical and non-surgical settings,^{13,14} these often include heterogeneous cohorts of patients. Therefore, we aimed to evaluate the safety and efficacy of perioperative TXA (any dose or route) focusing only on patients undergoing major non-cardiac emergency or elective vascular surgery.

Methods

This systematic review followed a protocol that was designed and reported according to Preferred Reporting Items for Systematic Reviews and Meta-Analyses (PRISMA) guidelines (see Appendix 1 online at www.jvsgbi.com).¹⁵ The protocol is registered on PROSPERO (CRD42023427282). Ethics committee approval was not required as this was a synthesis of previously published literature.

Information sources and search strategy

Relevant bibliographic databases including PubMed, EMBASE (via OVID), MEDLINE and Transfusion Evidence Library, were searched. The full search strategy (see Appendix 2 online at www.jvsgbi.com) included terms relating to or describing exposure (TXA), outcome (thrombosis, bleeding) and the study population (adults undergoing major non-cardiac vascular surgery). There was no restriction on the date of publication and only peer-reviewed RCTs published in academic journals were included.

Inclusion and exclusion criteria

Eligible RCTs included adult participants (as defined by the study authors) undergoing major non-cardiac vascular surgery and receiving either TXA (any route, dose and formulation) or placebo/no TXA during the perioperative period. Major non-cardiac vascular surgery was defined as including the following elective, urgent and emergency procedures: open or endovascular aneurysm repair of the abdominal aorta (AAA repair or EVAR), thoracic aorta, thoraco-abdominal aorta or lower limb artery; open or endovascular repair of aorta (thoracic, abdominal, thoracoabdominal) dissection; infrainguinal lower limb bypass surgery (including open and hybrid operations); major lower limb amputation (below, above and through knee amputations); carotid endarterectomy. Cohort studies (prospective and retrospective), case-control studies, letters, editorials, commentaries and case reports were excluded.

Outcomes

The primary outcome of interest was the risk of perioperative arterial and venous thromboembolic complications, which we defined as a composite of myocardial infarction, non-haemorrhagic stroke, peripheral arterial thrombosis and symptomatic proximal venous thromboembolism. Secondary outcomes that were analysed are as follows: critical bleeding (composite of lifethreatening, critical and major organ bleed), measured perioperative blood loss, risk of receiving allogeneic blood transfusion in the perioperative period and in-hospital mortality.

Given the concerns about thromboembolism associated with TXA in vascular surgery, safety was designated as the primary outcome for this review. Efficacy was chosen as the secondary outcome as it is safety concerns, rather than doubts about efficacy, that we hypothesise have deterred the use of TXA in vascular surgery.

Study selection and data collection process

Data from eligible studies were extracted using a pre-piloted spreadsheet. Extracted data included the number of participants in each study, participant characteristics, study duration, type of vascular operation undertaken, route and method of TXA and control administration, transfusion thresholds utilised (if stated), whether any additional cointerventions were used (eg, cell salvage), as well as information relating to quality assessment. This was done in duplicate by two authors (KA and LC) and disagreements were reviewed by a third author (KB or AS). For studies that did not report data on certain outcomes of interest, the relevant authors were contacted.

Risk of bias assessment

The Cochrane Collaboration's domain-based risk of bias (ROB) tool¹⁶ was used to assess risk of bias in each included study. Each of these parameters was evaluated for each study and scored on a 3-point scale corresponding to a low, unclear or high risk of bias. This was also done in parallel by two reviewers (KA and LC) and disagreements were resolved by a third author (KB or AS).

Data synthesis

Data were analysed using Cochrane Collaboration's Review Manager 5 (REVMAN 5) software. Wherever possible, a randomeffect model was used to synthesise data about primary and secondary outcomes. Measures of effect were reported as relative risk (RR) with 95% confidence intervals (CIs). Each outcome was evaluated as relevant deviations from the control group. When comparing end points of interest, statistical significance was established at a threshold of p<0.05. Heterogeneity between studies was assessed with the use of I².¹⁷ Any outcome data that could not be combined into a meta-analysis were synthesised narratively instead.

Assessment of the certainty of the evidence

The GRADE (Grades of Recommendation, Assessment,



Development, and Evaluation)¹⁸ approach was used to assess the overall quality of evidence for the key outcomes of thromboembolic complications and critical bleeding.

Results

Study selection

Our electronic search identified 1,989 relevant publications. Following title and abstract screening, eight articles were shortlisted. Subsequently, full texts were reviewed and three RCTs^{8,11,19} were ultimately included in this review (Figure 1). The reasons for excluding the five studies were: incorrect study design (ie, retrospective studies) (n=2), full texts in non-English language (n=2) and ongoing clinical trial (n=1). Details of the included RCTs are shown in Table 1. We found one ongoing RCT (NCT04803747) that is planned to complete recruitment in 2024.

Study characteristics

The three included RCTs enrolled a total of 1,560 patients across 24 different countries (Australia, Austria, Belgium, Brazil, Canada, Chile, China, Denmark, France, Germany, Hong Kong, India, Italy, Japan, Malaysia, Netherlands, New Zealand, Pakistan, Poland,

Figure 2 Risk of Bias (ROB) assessment based on the Cochrane Risk of Bias Tool. Each study was evaluated against different parameters and scored based on a 3-point scale corresponding to a low (green), unclear (yellow), or high risk of bias (red).



Russia, South Africa, Spain, UK and USA). There was one large multicentre international RCT which provided data on 1,360 participants who underwent vascular surgery.⁸ In comparison, the other two RCTs included 100 participants each.^{11,19}

In two RCTs the patients were administered intravenous TXA at a loading dose of 1 g preoperatively along with a continuous infusion intraoperatively or a bolus at the end. The third trial administered 750 mg oral TXA daily for a month after surgery (Table 1). Two of the RCTs only included patients undergoing elective AAA repair¹¹ and elective EVAR¹⁹ and the other RCT included patients undergoing both elective and emergency vascular surgery but data on specific types of surgery were not available.⁸

Risk of bias assessment

Two of the RCTs had a low risk of bias across all domains (Figure 2)¹¹ and one RCT was judged to be at high overall risk of bias due to concerns regarding allocation concealment and participant and assessor blinding.¹⁹

Primary outcome

Thromboembolic risk

Two studies (1,460 participants) reported data on the incidence of thromboembolic events in patients receiving intravenous TXA.^{8,11} Overall, when the data were pooled, there was no evidence of an effect of intravenous TXA on the incidence of thromboembolic events (RR 1.10, 95% CI 0.89 to 1.36) (Figure 3). On the basis of the GRADE framework, this finding was judged to be low certainty evidence (Table 2). One study (85 participants) evaluating oral TXA observed no thrombotic events during the study period.¹⁹ Given the variability in the methodology (ie, intravenous vs oral TXA), the study reporting results from oral TXA use was not included in the meta-analysis.

Secondary outcomes

Critical bleeding

Two studies (1,460 participants) reported data on the incidence of critical bleeding in patients receiving intravenous TXA. Overall, there was no evidence of an effect of intravenous TXA on the incidence of critical bleeding (RR 0.85, 95% CI 0.65 to 1.11, low certainty of evidence) (Figure 4).

Blood loss

Only one study reported information on blood loss during the first 4 and 24 hours postoperatively.¹¹ During the first 4 hours, participants who received TXA lost a median (IQR) of 60 (40–80) mL blood compared with participants who received placebo who lost a median (IQR) of 100 (60–140) mL blood (Cliff's delta –0.41 (–0.19 to –0.59), p<0.001). Similarly, during the 24 hours post-surgery, participants receiving TXA continued to have a significantly lower volume of blood loss compared with those who received placebo (Cliff's delta –0.37 (–0.14 to –0.55), p=0.002) (Table 3).

Study author	Methods	Participants	Intervention	Comparator(s)	Primary outcome	Secondary outcome(s)
Devereaux <i>et al</i> , 2022 ⁸	International multicentre parallel RCT across 114 sites	Age >45 years, undergoing inpatient, non- cardiac, emergency or elective surgery and at risk for bleeding and cardiovascular complications* (n=1360)**	IV TXA 1 g as bolus or 10-min infusion One dose at the start of surgery (20 minutes before incision) and one dose at the end	IV 0.9% saline as bolus or 10-min infusion One dose at the start of surgery (20 minutes before incision) and one dose at the end	Composite of life- threatening, major and critical organ bleeding at 30 days after randomisation Composite of MINS, non-haemorrhagic stroke, peripheral arterial thrombosis and symptomatic proximal venous thromboembolism at 30 days after randomisation	 Bleeding independently associated with mortality after non-cardiac surgery Life-threatening bleeding Major bleeding Critical organ bleeding; MINS MINS not fulfilling the universal definition of myocardial infarction Composite of vascular death, bleeding (ie, non-fatal life- threatening, major or critical organ), MINS, stroke, peripheral arterial thrombosis and symptomatic proximal venous thromboembolism (ie, a net risk-benefit outcome)
Imaeda <i>et al</i> , 2022 ¹⁹	Single-centre parallel RCT (Japan)	Adult patients undergoing EVAR (n=100)	Oral TXA 750 mg every day (250 mg after each meal) for 30 days after surgery	No TXA	Incidence of EL2 at 1 week and 1 month after EVAR	 (1) Changes in the aneurysm diameter at 1 week, 1 month, 6 months and 1 year after EVAR (2) Results of the blood coagulation/ fibrinolysis test at 3 days, 7 days and 1 month after EVAR (3) Rate of change in aneurysm diameter from baseline to 6 months and 1 year
Monaco <i>et al</i> , 2020 ¹¹	Single-centre parallel RCT (Italy)	Age >50 years, undergoing elective open AAA surgical repair (n=100)	IV TXA loading dose of 500 mg diluted in 100 mL slowly infused before surgery Continuous undiluted infusion rate of 250 mg/hour (2.5 mL/hour) during surgery	IV 0.9% saline loading dose of 500 mg diluted in 100 mL slowly infused before surgery Continuous undiluted infusion rate of 250 mg/hour (2.5 mL/hour) during surgery	Intraoperative blood loss (sum of blood volume aspired during surgery and blood volume absorbed in gauzes (at the end of the surgery, all gauzes used were weighted and net weight of the gauzes was subtracted from the total weight))	 (1) RBC transfusion requirements (2) Thromboembolic events (including acute MI, PE, bowel infarction or seizures) up to 28 days after surgery (3) Mortality: 28-day, 1-year
Ongoing NCT04803747	Multicentre RCT	Age >18 years undergoing major non-cardiac surgery with estimated risk >5% of requiring RBC and duration of surgery >3 hours (n=8320)	IV TXA 1 g bolus. One dose within 10 minutes of skin incision and one dose at 2–4 hours of surgery or prior to skin closure	IV 0.9% saline as bolus. One dose within 10 minutes of skin incision and one dose at 2–4 hours of surgery or prior to skin closure	RBC transfusion requirements Incidence of DVT or PE within 3 months of surgery	 Number of RBC units transfused In-hospital MI, stroke, DVT or PE Hospital length of stay ICU admission 90-day mortality Days at home to day 30

*Risk of bleeding and cardiovascular complications defined as known atherosclerotic disease, undergoing major surgery, age ≥70 years and serum creatinine level >175 µmol/L. **Study included a total of 9,535 individuals undergoing several different higher risk surgeries. 1,360 of these participants underwent vascular surgeries. Details of the specific type of vascular surgery and the nature of the vascular surgery (emergency vs elective) are unknown.

AAA, abdominal aortic aneurysm; DVT, deep vein thrombosis; EL2, type 2 endoleak; EVAR, endovascular aneurysm repair; ICU, intensive care unit; IV, intravenous; MI, myocardial infarction; MINS, myocardial injury after non-cardiac surgery; PE, pulmonary embolism; RBC, red blood cell; RCT, randomised controlled trial; TXA, tranexamic acid.

Figure 3 Forest plot of the effect of tranexamic acid on the primary composite outcome: thromboembolic risk (composite of myocardial infarction, non-haemorrhagic stroke, peripheral arterial thrombosis and symptomatic proximal venous thromboembolism).

Tranexamic acid			Control Risk Ratio			Risk Ratio	Risk Ratio				
Study or Subgroup Events Total			Events Total Weight M-H, Random, 95%			M-H, Random, 95% Cl	CI M-H, Ran <u>dom,</u> 95% CI				
Devereaux 2022 Monaco 2019	140 2	684 50	126 2	676 50	98.8% 1.2%	1.10 (0.88, 1.36) 1.00 (0.15, 6.82)			-		
Total (95% CI) Total events	142	734	128	726	100.0%	1.10 (0.89, 1.36) -			•		
Heterogeneity: Tau ² = 0.00; Chi ² = 0.01, df = 1 (P=0.92): I ² = 0%								0.2	1 5	20	
Test for overall effect: $Z = 0.84$ (P=0.40)								Favours control	Favours tranexar	nic acid	

M-H, Mantel-Haenszel

Table 2 GRADE assessment of key clinical outcomes.

Certainty assessment								Summary of findings				
							Study eve	ent rates (%)		Antio absolu	cipated te effects	
Participants (studies) Follow-up	Risk of bias	Inconsistency	Indirectness	Imprecision	Publication bias	Overall certainty of evidence	With placebo/ control	With tranexamic acid	Relative effect (95% CI)	Risk with placebo/ control	Risk difference with tran- examic acid	
Thrombo	Thromboembolic complications											
1460 (2 RCTs)	Not serious	Serious*	Not serious	Serious**	None	⊕⊕○○ Low	128/726 (17.6%)	142/734 (19.3%)	RR 1.10 (0.89 to 1.36)	176 per 1000	18 more per 1000 (from 19 fewer to 63 more)	
Critical b	Critical bleeding											
1499 (2 RCTs)	Not serious	Serious*	Not serious	Serious**	None	⊕⊕⊖⊖ Low	104/750 (13.9%)	88/749 (11.7%)	RR 0.85 (0.65 to 1.11)	139 per 1000	21 fewer per 1000 (from 49 fewer to 15 more)	

*Downgraded due to differences in direction of point estimate **Wide confidence intervals (CIs). RCT, randomised controlled trial

Figure 4 Forest plot of the effect of tranexamic acid on the secondary composite outcome: critical bleeding (composite of life-threatening, critical and major organ bleed).

Study or Subgroup	Tranexam Events	nic acid Total	Control Events	Total	Weight	Risk Ratio M-H, Random, 95% Cl	Ris M-H, Rar	sk Ratio ndom, 95% Cl	
Devereaux 2022 Monaco 2019	88 0	699 50	103 1	700 50	99.3% 0.7%	0.86 (0.66, 1.11) 0.33 (0.01, 7.99)			
Total (95% CI) Total events Heterogeneity: $Tau^2 = 0.00$;	88 Chi ² = 0.34	749	104 =0.56): I ² =	750 0%	100.0%	0.85 (0.65, 1.11) 0.01	0.1 Favours tranexamic acid	1 10 Favours control	100

M-H, Mantel-Haenszel

Outcome	Author, year	Number	TXA (events/total) (%)	Control (events/total) (%)	RR (95% CI)
Perioperative RBC transfusion requirements	Monaco <i>et al,</i> 2020 ¹¹	100	7/50 (14%)	12/50 (24%)	0.66 (0.11 to 3.95)
Composite clinically important bleeding	Monaco <i>et al</i> , 2020 ¹¹ and Devereaux <i>et al</i> , 2022 ⁸	1499	88/749 (12%)	104/750 (14%)	0.85 (0.65 to 1.11)
Outcome	Author, year	Number	TXA (n=50)	Control (n=50)	Cliff's delta (95% Cl)
Blood loss 0–4 hours after surgery, median (IQR), mL	Monaco <i>et al</i> , 2020 ¹¹	100	60 (40-80)	100 (60–140)	-0.41 (-0.19 to -0.59)
				075 (100,005)	

Table 3 Summary of results of secondary outcome

CI, confidence interval; IQR, interquartile range; RBC, red blood cell; RR, relative risk; TXA, tranexamic acid.

Risk of receiving allogeneic RBC transfusion

One study (n=100) reported data on perioperative blood transfusion requirements and found weak evidence of an effect of TXA on reducing transfusion requirements (RR 0.66, 95% CI 0.11 to 3.95) (Table 3). Devereaux et al8 reported on allogeneic RBC transfusion across all included surgical specialties (HR 0.77, 95% CI 0.68 to 0.88) but did not provide transfusion data specific to vascular surgery. We contacted them for additional data regarding the vascular subgroup but they were unable to share it at this time.

In-hospital mortality

One study (n=100) provided data on in-hospital mortality and did not report deaths in either study arm.

Oral TXA

Imaeda *et al* evaluated the use of oral TXA for the prevention of type II endoleak in adults undergoing EVAR (n=100).¹⁹ Acknowledging the heterogeneity in the administration of intravenous and oral TXA as well as the variability in reported outcomes, we have narratively summarised the findings of this trial. The study reported that seven days post EVAR, type II endoleak was observed in 14 patients (34.1%) in the TXA group and in seven patients (15.9%) in the non-TXA group. At 1 month follow-up post EVAR, 12 patients (29.3%) in the TXA group and six patients (13.6%) in the non-TXA group were found to have type II endoleaks. There was no significant difference between the two groups in the incidence of type II endoleak (p=0.051 and p=0.08). No adverse events such as thrombus formation due to oral TXA were observed during the study period.

Discussion

The key findings of this systematic review are: (1) there is no

evidence of an effect of TXA on thromboembolic events or on critical bleeding in patients undergoing major non-cardiac vascular surgery (low certainty of evidence); (2) while TXA appears to reduce early postoperative bleeding, the clinical relevance of this remains unclear, particularly in an era of patient blood management interventions²⁰ such as cell salvage, restrictive transfusion and minimally invasive surgical techniques; and (3) the current evidence base is limited by the small number of RCTs and wide Cls around the effect estimates, which could encompass clinically important differences.

Both bleeding and thrombosis in patients undergoing vascular surgery are associated with poor outcomes and the underlying pathophysiology and mechanisms are multifactorial. Perioperative risk factors for bleeding include systemic anticoagulation with heparin (to prevent graft thrombosis or clot extension), preoperative use of antiplatelets and/or anticoagulants, intraoperative hypothermia, cross-clamp position, haemodilution and consumption coagulopathy during ongoing blood loss.²¹ Similarly, a procoagulant state has been observed in AAA repair with increased thrombin generation and inhibition, which may lead to microvascular and macrovascular thrombosis. This can eventually manifest as myocardial infarction, graft thrombosis, multi-organ failure and venous thromboembolism.²² The presence of hypercoagulable comorbidities is associated with an increased risk of developing venous thromboembolism and graft thrombosis in patients undergoing vascular surgery.^{23,24}

The incidence of perioperative RBC transfusion in patients undergoing vascular surgery ranges from 0% to 25%.^{25,26} Both excessive bleeding and perioperative RBC transfusion are associated with increased postoperative morbidity and mortality.^{25,27} Certain patient groups such as those with preoperative anaemia, Jehovah's Witnesses or risk factors for bleeding (eg, anticoagulants) may benefit from TXA, but further research is needed.

Operations at risk of moderate-to-severe blood loss (at least 500 mL) where the use of TXA is recommended by national quidelines include open AAA repair, complex lower limb bypass surgery and major lower limb amputation.²⁸ However, a survey of anaesthetists in Australia and New Zealand found considerable variability in the use of TXA in vascular procedures. While 3% of vascular anaesthetists routinely administered TXA, 40% did not administer it at all and 20% and 37% reported administering TXA selectively and on surgical request, respectively.²⁹ The reasons for this are unclear but may relate to concerns regarding thrombosis. It is worth noting that, in the study by Devereaux et al,⁸ TXA did not meet the non-inferiority margin for the thromboembolic composite safety outcome, although some have argued that the upper bound of the CI (1.14) just surpassed the non-inferiority margin (1.125) and that it is likely that the true effect lies below this margin.³⁰ Others have expressed concerns regarding widespread prophylactic use in the absence of established fibrinolytic bleeding, and that a more individualised approach may be needed.^{16,31} It is unclear whether similar patterns exist in the UK or elsewhere and how they may differ between elective and emergency surgery cases.

Our findings are broadly in agreement with recent work by Tsan et al,¹³ although they did not include data from Devereaux et al⁸ in their subgroup analysis for vascular surgery. The strength of this review is the strict methodological process. We followed Cochrane Collaboration. PRISMA and GRADE recommendations. We conducted a comprehensive search of multiple databases and clinical trial registries to ensure all relevant studies would be captured. All screening, data extraction and risk of bias assessments were done in duplicate. Limitations of this review stem from the limited number of studies included. There were only three eligible studies and only two were included in the meta-analysis given methodological and reporting differences between trials. Once available, we aim to update this review with data from an ongoing study (NCT04803747) to reduce the uncertainty on the benefits and risks of TXA in patients undergoing major non-cardiac vascular (or other) surgery.

The 2023 NHS Blood and Transfusion (NHSBT) National Comparative Audit in the UK found that only 900 out of 1,336 surgical patients were given TXA. Around one-third (32.6%) of patients who were eligibile for it did not receive it.³² This prompts consideration of factors that are preventing clinicians from using TXA, even when it has been shown to reduce major bleeding by 25% without increasing the risk of thromboembolic events. Additionally, the audit also reported "the low use of TXA in vascular surgery noteworthy (13.5%)". We hypothesise that this may be explained by the paucity of evidence that we have identified in this systematic review. Evidently, the currently available data on the use of TXA in vascular surgery are insufficient to definitively guide clinical practice either for or against its use.

The publication of the Infected Blood Inquiry report along with

- We found no evidence of an effect of TXA on thromboembolic complications or on critical bleeding.
 However, the certainty of the evidence was judged to be low.
- TXA reduces early postoperative bleeding but the clinical relevance of this remains unclear. There was no evidence of an effect on red blood cell transfusion use.
- High-quality research is needed to determine the riskbenefit balance of TXA in vascular surgery.

recent national blood shortages has again re-ignited efforts to promote the use of TXA. Example recommendations include incorporating TXA in preoperative WHO checklists.^{33,34} It is likely that a key reason TXA is not widely used in vascular surgery is due to safety concerns rather than issues of oversight during procedures. Hence, until these concerns are addressed, vascular surgeons and anaesthetists are unlikely to start using TXA merely because it is included in preoperative checklists.

At the moment there are no ongoing trials studying TXA solely in patients undergoing vascular surgery. However, if such a trial were to be conducted, it would need to include patients undergoing open procedures at high risk of blood loss. The end points of this trial should assess both safety (incidence of thromboembolic events) and efficacy (effects on major bleeding and RBC transfusion requirements). Future trials should also consider 'enrichment' strategies by enrolling patients at high risk of experiencing the outcomes of interest (eg, major bleeding, thrombosis) such as those on antiplatelet and anticoagulant therapy, scheduled to undergo complex surgery and likely to require prolonged aortic cross-clamping duration, and those with multiple comorbidities. Although a pragmatic TXA dose of 1 g is often used, the optimal dose, route and timing is still a topic of active research. For the time being, our data may provide some reassurance to vascular anaesthetists and surgeons with regard to the risk of thromboembolic complications, but further research is needed.

Conflict of Interest: AS is an Editor of *Anaesthesia* and has received consultancy fees from Pharmacosmos UK outside of the submitted work.

Funding: None.

Author contributions: KA: study design; data extraction and analysis; preparation of figures and composition of the initial and final manuscript. LC: data extraction and analysis; composition of the final manuscript. KB: study design; data analysis; composition of the final manuscript; content expertise and co-supervision. AS: study design; data analysis; preparation of figures; composition of the final manuscript; overall supervision of this work.

Acknowledgement: We would like to thank Dr Paula Dhiman for providing statistical advice during the preparation of this manuscript.

Reviewer acknowledgement: *JVSGBI* thanks Sandip Nandhra, Northern Vascular Centre, Freeman Hospital and Rob Sayers, Glenfield Hospital Leicester, for their contribution to the peer review of this work.

References

- Patel PA, Wyrobek JA, Butwick AJ, et al. Update on applications and limitations of perioperative tranexamic acid. Anesth Analg 2022;135(3):460– 73. http://dx.doi.org/10.1213/ANE.00000000006039
- CRASH-2 Trial Collaborators, Shakur H, Roberts I, Bautista R, *et al.* Effects of tranexamic acid on death, vascular occlusive events, and blood transfusion in trauma patients with significant haemorrhage (CRASH-2): a randomised, placebo-controlled trial. *Lancet* 2010;**376**(9734):23–32. http://dx.doi.org/10.1016/S0140-6736(10)60835-5
- WOMAN Trial Collaborators. Effect of early tranexamic acid administration on mortality, hysterectomy, and other morbidities in women with post-partum haemorrhage (WOMAN): an international, randomised, double-blind, placebocontrolled trial. *Lancet* 2017;389:2105–16. https://doi.org/10.1016/S0140-6736(17)30638-4
- CRASH-3 Collaborators. Effects of tranexamic acid on death, disability, vascular occlusive events and other morbidities in patients with acute traumatic brain injury (CRASH-3): a randomised, placebo-controlled trial. *Lancet* 2019;**394**:1713–23. https://doi.org/10.1016/S0140-6736(19)32233-0
- Myles PS, Smith JA, Forbes A, *et al.* Tranexamic acid in patients undergoing coronary-artery surgery. *N Engl J Med* 2017;**376**(2):136–48. http://dx.doi.org/10.1056/nejmoa1606424
- Siemens K, Donnelly P, Hunt BJ, Carter MJ, Murdoch IA, Tibby SM. Evaluating the impact of cardiopulmonary bypass priming fluids on bleeding after pediatric cardiac surgery: a systematic review and meta-analysis. *J Cardiothorac Vasc Anesth* 2022;**36**(6):1584–94. http://dx.doi.org/10.1053/j.jvca.2021.11.031
- Sentilhes L, Winer N, Azria E, et al. Tranexamic acid for the prevention of blood loss after vaginal delivery. N Engl J Med 2018;379(8):731–42. http://dx.doi.org/10.1056/NEJMoa1800942
- Devereaux PJ, Marcucci M, Painter TW, et al. Tranexamic acid in patients undergoing noncardiac surgery. N Engl J Med 2022;386(21):1986–97. http://dx.doi.org/10.1056/NEJMoa2201171
- Piggott T, Nowak A, Brignardello-Petersen R, et al. Global status of essential medicine selection: a systematic comparison of national essential medicine lists with recommendations by WHO. BMJ Open 2022;12(2):e053349. http://dx.doi.org/10.1136/bmjopen-2021-053349
- HALT-IT Trial Collaborators. Effects of a high-dose 24-h infusion of tranexamic acid on death and thromboembolic events in patients with acute gastrointestinal bleeding (HALT-IT): an international randomised, double-blind, placebo-controlled trial. *Lancet* 2020;**395**(10241):1927–36. http://dx.doi.org/10.1016/S0140-6736(20)30848-5
- Monaco F, Nardelli P, Pasin L, et al. Tranexamic acid in open aortic aneurysm surgery: a randomised clinical trial. Br J Anaesth 2020;**124**(1):35–43. http://dx.doi.org/10.1016/j.bja.2019.08.028
- Gerstein NS, Kelly SP, Brierley JK. Yet another tranexamic acid-related thrombotic complication. J Cardiothorac Vasc Anesth 2016;30(3):e21-2. http://dx.doi.org/10.1053/j.jvca.2015.11.023
- Tsan SEH, Viknaswaran NL, Cheong CC, et al. Prophylactic intravenous tranexamic acid and thromboembolism in non-cardiac surgery: a systematic review, meta-analysis and trial sequential analysis. Anaesthesia 2023;**78**(9): 1153–61. http://dx.doi.org/10.1111/anae.16058
- Taeuber I, Weibel S, Herrmann E, et al. Association of intravenous tranexamic acid with thromboembolic events and mortality: a systematic review, metaanalysis, and meta-regression. JAMA Surg 2021;156(6):e210884. http://dx.doi.org/10.1001/jamasurg.2021.0884
- Page MJ, McKenzie JE, Bossuyt PM, *et al.* The PRISMA 2020 statement: an updated guideline for reporting systematic reviews. *BMJ* 2021;**372**:n71. http://dx.doi.org/10.1136/bmj.n71
- Higgins JPT, Altman DG, Gøtzsche PC, *et al.* The Cochrane Collaboration's tool for assessing risk of bias in randomised trials. *BMJ* 2011;**343**:d5928. http://dx.doi.org/10.1136/bmj.d5928
- Higgins JPT, Thompson SG, Deeks JJ, Altman DG. Measuring inconsistency in meta-analyses. *BMJ* 2003;**327**(7414):557–60. http://dx.doi.org/10.1136/bmj.327.7414.557
- Atkins D, Best D, Briss PA. GRADE Working Group. Grading quality of evidence and strength of recommendations. *BMJ* 2004;**328**:1490. http://dx.doi.org/10.1136/ bmj.328.7454.1490

- Imaeda Y, Ishibashi H, Orimoto Y, Maruyama Y, Mitsuoka H, Arima T. Randomized controlled trial of oral tranexamic acid intervention for the prevention of type II endoleak after endovascular abdominal aneurysm repair. *Ann Thorac Cardiovasc Surg* 2022;28(4):286–92. http://dx.doi.org/10.5761/atcs.oa.22-00100
- Shander A, Hardy J-F, Ozawa S, et al. A global definition of patient blood management. Anesth Analg 2022;**135**(3):476–88. http://dx.doi.org/10.1213/ANE.00000000005873
- Shah A, Palmer AJR, Klein AA. Strategies to minimize intraoperative blood loss during major surgery: minimizing intraoperative blood loss during major surgery. *Br J Surg* 2020;**107**(2):e26–38. http://dx.doi.org/10.1002/bjs.11393
- Adam DJ, Ludlam CA, Ruckley CV, Bradbury AW. Coagulation and fibrinolysis in patients undergoing operation for ruptured and nonruptured infrarenal abdominal aortic aneurysms. *J Vasc Surg* 1999;**30**(4):641–50. http://dx.doi.org/10.1016/s0741-5214(99)70103-5
- Matthay ZA, Flanagan CP, Sanders K, *et al.* Risk factors for venous thromboembolism after vascular surgery and implications for chemoprophylaxis strategies. *J Vasc Surg Venous Lymphat Disord* 2022; 10(3):585–593.e2. http://dx.doi.org/10.1016/j.jvsv.2021.10.001
- Baril DT, Patel VI, Judelson DR, et al. Outcomes of lower extremity bypass performed for acute limb ischemia. J Vasc Surg 2013;58(4):949–56. http://dx.doi.org/10.1016/j.jvs.2013.04.036
- Obi AT, Park YJ, Bove P, *et al.* The association of perioperative transfusion with 30-day morbidity and mortality in patients undergoing major vascular surgery. *J Vasc Surg* 2015;**61**(4):1000–1009.e1. http://dx.doi.org/10.1016/j.jvs.2014.10.106
- Stoneham MD, Barbosa A, Maher K, Douglass P, Desborough MJR, Von Kier S. Intraoperative cell salvage using swab wash and serial thromboelastography in elective abdominal aortic aneurysm surgery involving massive blood loss. *Br J Haematol* 2023;**200**(5):652–9. http://dx.doi.org/10.1111/bjh.18523
- Osborne Z, Hanson K, Brooke BS, *et al.* Variation in transfusion practices and the association with perioperative adverse events in patients undergoing open abdominal aortic aneurysm repair and lower extremity arterial bypass in the vascular quality initiative. *Ann Vasc Surg* 2018;**46**:1–16. http://dx.doi.org/10.1016/j.avsg.2017.06.154
- UK Royal Colleges Tranexamic Acid in Surgery Implementation Group, Grocott MPW, Murphy M, Roberts I, Sayers R, Toh C-H. Tranexamic acid for safer surgery: the time is now. *Br J Anaesth* 2022;**129**(4):459–61. http://dx.doi.org/10.1016/j.bja.2022.06.024
- Painter TW, McIlroy D, Myles PS, Leslie K. A survey of anaesthetists' use of tranexamic acid in noncardiac surgery. *Anaesth Intensive Care* 2019; 47(1):76–84. http://dx.doi.org/10.1177/0310057X18811977
- UK Royal Colleges Tranexamic Acid in Surgery Implementation Group, Grocott MPW, Murphy M, Roberts I, Sayers R, Toh C-H. Tranexamic acid for safer surgery: does the evidence support preventative use? Response to Br J Anaesth 2023;**130**:e23–4. Br J Anaesth 2023;**130**(2):e195–6. http://dx.doi.org/10.1016/j.bja.2022.11.010
- Jones AL. Tranexamic acid for safer surgery: does the evidence support preventative use? Comment on Br J Anaesth 2022;129:459–61. Br J Anaesth 2023;130(1):e23–4. http://dx.doi.org/10.1016/j.bja.2022.10.016
- NHS Blood and Transplant (NHSBT). 2023 National Comparative audit of NICE Quality Standard QS138 [Internet]. Available from: https://hospital.blood.co.uk/audits/national-comparative-audit/reportsgrouped-by-year/2023-national-comparative-audit-of-nice-quality-standard-qs 138/ [cited 2024 May 21].
- Infected Blood Inquiry. Witness Statement WITN731001 [Internet]. Available from: https://www.infectedbloodinquiry.org.uk/sites/default/files/5001-5191%20Part%203%20d/5001-5191%20Part%203%20d/WITN7310001%20
 -%20Written%20statement%20of%20Professor%20Ian%20Roberts%20-%2016%20Nov%202022.pdf [Accessed 05 August 2024]
- Roberts I, Murphy MF, Moonesinghe R, et al. Wider use of tranexamic acid to reduce surgical bleeding could benefit patients and health systems. BMJ 2024;385:e07944. http://doi.org/10.1136/bmj-2024-079444

PRISI	MA 2	020 Checklist	
Section and Topic	ltem #	Checklist item	Reported on page no.
TITLE			
Title	1	Identify the report as a systematic review.	Title page
ABSTRACT			
Abstract	2	See the PRISMA 2020 for Abstracts checklist.	3,4
INTRODUCTION			
Rationale	3	Describe the rationale for the review in the context of existing knowledge.	5
Objectives	4	Provide an explicit statement of the objective(s) or question(s) the review addresses.	5
METHODS			
Eligibility criteria	5	Specify the inclusion and exclusion criteria for the review and how studies were grouped for the syntheses.	9
Information sources	9	Specify all databases, registers, websites, organisations, reference lists and other sources searched or consulted to identify studies. Specify the date when each source was last searched or consulted.	9
Search strategy	7	Present the full search strategies for all databases, registers and websites, including any filters and limits used.	Supp. Appendix 2
Selection process	8	Specify the methods used to decide whether a study met the inclusion criteria of the review, including how many reviewers screened each record and each record and each report retrieved, whether they worked independently, and if applicable, details of automation tools used in the process.	7
Data collection process	6	Specify the methods used to collect data from reports, including how many reviewers collected data from each report, whether they worked independently, any processes for obtaining or confirming data from study investigators, and if applicable, details of automation tools used in the process.	7
Data items	10a	List and define all outcomes for which data were sought. Specify whether all results that were compatible with each outcome domain in each study were sought (e.g. for all measures, time points, analyses), and if not, the methods used to decide which results to collect.	6, 7
	10b	List and define all other variables for which data were sought (e.g. participant and intervention characteristics, funding sources). Describe any assumptions made about any missing or unclear information.	6, 7
Study risk of bias assessment	11	Specify the methods used to assess risk of bias in the included studies, including details of the tool(s) used, how many reviewers assessed each study and whether they worked independently, and if applicable, details of automation tools used in the process.	7
Effect measures	12	Specify for each outcome the effect measure(s) (e.g. risk ratio, mean difference) used in the synthesis or presentation of results.	7,8
Synthesis methods	13a	Describe the processes used to decide which studies were eligible for each synthesis (e.g. tabulating the study intervention characteristics and comparing against the planned groups for each synthesis (item #5)).	Fig 1., Table 1
	13b	Describe any methods required to prepare the data for presentation or synthesis, such as handling of missing summary statistics, or data conversions.	7,8
	13c	Describe any methods used to tabulate or visually display results of individual studies and syntheses.	7,
	13d	Describe any methods used to synthesize results and provide a rationale for the choice(s). If meta-analysis was performed, describe the model(s), method(s) to identify the presence and extent of statistical heterogeneity, and software package(s) used.	7
	13e	Describe any methods used to explore possible causes of heterogeneity among study results (e.g. subgroup analysis, meta-regression).	-
	13f	Describe any sensitivity analyses conducted to assess robustness of the synthesized results.	
Reporting bias assessment	14	Describe any methods used to assess risk of bias due to missing results in a synthesis (arising from reporting biases).	7,8
Certainty	15	Describe any methods used to assess certainty (or confidence) in the body of evidence for an outcome.	8

Appendix 1 PRISMA 2020 checklist.

Appendix 2 Search strategy.

Date: 8th April, 2023

CENTRAL (The Cochrane Library Issue 6, 2021)

Number	Search Term	Results
#1	vascular surgical procedures	3234
#2	Vascular surgery	15842
#3	Endarterectomy	2235
#4	Endovascular*	5646
#5	Axillofemoral bypass grafting	9
#6	Embolectom*	161
#7	Limb salvage	538
#8	Amputation	3529
#9	Thrombectomy	2048
#10	Vascular grafting	1253
#11	Blood vessel prosthesis implantation	628
#12	Arteriovenous shunt	474
#13	Vascular disease	26105
#14	Aneurysm	4660
#15	Aortic Disease	4267
#16	Aortic	11340
#17	Arterial Occlusive Disease	2182
#18	Arteriovenous Malformation	252
#19	Diabetic Angiopath*	1243
#20	Peripheral vascular disease	6183
#21	Spinal cord vascular disease	228
#22	Vascular system injur*	1413
#23	Venous insufficiency	1906
#24	#1 or #2 or #3 or #4 or #5 or #6 or #7 or #8 or #9 or #10 or #11 or #12 or	57505
	#13 or #14 or #15 or #16 or #17 or #18 or #19 or #20 or #21 or #22 or #23	
#25	Tranexamic acid	3654
#26	TXA	1192
#27	Antifibrinolytic	1464
#28	Antiplasmin	362
#29	Plasmin inhibitor	259